Author's response to reviews

Title: Seroprevalence and Risk Factors for Toxoplasma Infection among Pregnant Women in Aydin Province, Turkey

Authors:

Sema Ertug (sertug@adu.edu.tr)
Pinar Okyay (pinarokyay@adu.edu.tr)
Munevver Turkmen (turkmenm2000@yahoo.com)
Hasan Yuksel (hyuksel@adu.edu.tr)

Version: 3 Date: 8 June 2005

Author’s response to reviews: see over
June 8th, 2005

Dear Editor,

Thank you for your mail on June 7th, 2005, on our manuscript titled “Seroprevalence and Risk Factors for Toxoplasma Infection in Pregnancy, Aydin Province, Turkey”. We have revised our manuscript with respect to reviewer’s comments. Followings were done:

1. Point-by-point responses to the reviewers were prepared.
2. The whole text was controlled and corrected according to the style of the Journal.

Hope to hear you soon.

Yours sincerely,

Pinar Okyay
Corresponding author
Assistant Professor, Department of Public Health
Medical Faculty of Adnan Menderes University
Aydin, Turkey
Tel: +90.256.225 31 66 / 150
Fax: +90.256.212 01 46
E-mail: pinarokyay@adu.edu.tr
Our answers for the reviewer:

We have revised our manuscript with respect to reviewer’s comments and want to thank the reviewer for the precious contributions again.

Reviewer: Chia-Kwung Fan

1. We have already submitted “abstract” of the manuscript as in the previous submissions.

2. Mistyping as “1.69%” was corrected to “1.69%”.

3. In methods section, full names of the test were given when they first appeared in the text. Then, the abbreviations of them were used through the text.